Author's response to reviews

Title: Effectiveness of the Chest Pain Choice decision aid in emergency department patients with low risk chest pain: study protocol for a multicenter randomized trial

Authors:

Ryan T Anderson (anderson.ryan1@mayo.edu)
Victor M Montori (montori.victor@mayo.edu)
Nilay D Shah (shah.nilay@mayo.edu)
Henry H Ting (ting.henry@mayo.edu)
Laurie Pencille (pencille.laurie@mayo.edu)
Michele Demers (famille.leblanc.demers@gmail.com)
Jeffrey A Kline (Jefkline@iupui.edu)
Deborah B Diercks (dbdiercks@ucdavis.edu)
Judd E Hollander (Judd.Hollander@uphs.upenn.edu)
Carlos Torres (Torres.Carlos@mayo.edu)
Jason T Schaffer (jtschaff@iupui.edu)
Jeph Herrin (jeph.herrin@yale.edu)
Megan E Branda (branda.megan@mayo.edu)
Annie Leblanc (Lelanc.Annie@mayo.edu)
Erik P Hess (hess.erik@mayo.edu)

Version: 4
Date: 6 March 2014

Author's response to reviews:

I have made every effort to revise the manuscript based on the feedback provided below.

"Please modify your Authors' Contributions section to demonstrate that each author meets all three of the following criteria to qualify for authorship:

1. Substantial contributions to: the conception or design of the work; or the acquisition, analysis, or interpretation of data for the work; AND
2. Drafting the work or revising it critically for important intellectual content; AND
3. Final approval of the version to be published

Currently, RTA, MD, JH, MB and AL are missing point 2 of the above requirements. We suggest the following kind of format for this section:

“EP: conception and design, data collection and analysis, manuscript writing and final approval of the manuscript. SP: data collection and analysis, critical revision and final approval of the manuscript. MG: data collection and analysis, critical revision and final approval of the manuscript. AP: data collection and analysis and final approval of the manuscript. LS: data collection and analysis and final approval of the manuscript. MCL: data collection and analysis and final approval..."
of the manuscript. FB: conception and design, financial support, manuscript writing, final approval of manuscript. XH: conception and design, data collection and analysis, manuscript writing, final approval of the manuscript. All authors read and approved the final manuscript.”

Please let me know if there are additional revisions that need to be made.

yours,

Erik